Author’s response to reviews

Title: Relationship between Alterations of Urinary Microbiota and Cultured Negative Lower Urinary Tract Symptoms in Female Type 2 Diabetes Patients

Authors:

Jiawei Chen (miwaicjw@163.com)
Jie Zhao (zhaojie-0412@163.com)
Ying Cao (nfcy123@126.com)
Guihao Zhang (jyzhanggh@163.com)
Yang Chen (cyhm28@sina.com)
Jialei Zhong (15625171604@163.com)
Weina Huang (1287823961@qq.com)
Jiarong Zeng (15521266641@163.com)
Peng Wu (doctorwupeng@gmail.com)

Version: 3 Date: 31 Jul 2019

Author’s response to reviews:

1. Consent to participate
We note that you have included the following statement on Ethics approval and consent to participate statement of the Declarations: ‘…all co-authors consented to participate in the job in written format.’
Please note that the consent to participate refers to the study participants, and not the authors. Please amend the Ethics approval and consent to participate statement of the Declarations and confirm whether informed consent was obtained from all participants and clearly state this in your manuscript. Please specify whether the consent was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.
Answer: Changes have been made as suggested.
(Declarations section, line 521-523, page 24):
Ethics approval and consent to participate: This study was approved by the Ethical Committee of Southern Medical University; all participants were briefed on the procedure and asked to write informed consent.
2. Additional files
Please cite Table S2 in the main body of text.
Please rename the Figures contained within the Additional files to Figure S1, Figure S2, etc. to improve clarity. Please ensure that these additional figures are all cited in the main body of text.
Please note that any figures/tables/additional files that are not cited in the main body of text may be deleted by the production team.
Answer: Changes have been made as suggested.
(Declarations section, line 273-275, page 13):
Higher scores of AUA-SI (total score, 13.059±6.805 vs. 4.867±3.796; storage score, 7.824±4.825 vs. 2.600±2.230; emptying score, 5.235±5.118 vs. 2.267±3.369) were found in HH group (P < 0.05, Additional file 1: Table S2).

3. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Answer: Changes have been made as suggested.